Author's response to reviews

Title: Performance of capnometry in non-intubated infants in the Pediatric Intensive Care Unit

Authors:

Bria M Coates (b-coates@northwestern.edu)
Robin Chaize (rchaize@hotmail.com)
Denise M Goodman (dgoodman@luriechildrens.org)
Ranna A Rozenfeld (rozenfeld@luriechildrens.org)

Version: 2 Date: 12 November 2013

Author's response to reviews:

1. Acknowledgements section was included. There are no acknowledgements needed for this manuscript and no funding sources to include.
2. The table was added to the body of the manuscript.
3. The figures were changed to TIFF files and cropped.